Author's response to reviews

Title: A survey of trainee specialists experiences at the University of Cape Town (UCT): Impacts of race and gender

Authors:

Leslie London (leslie.london@uct.ac.za)
Sebastiana Kalula (sebastiana.kalula@uct.ac.za)
Bonga Xaba (XBXBON004@mail.uct.ac.za)

Version: 4 Date: 18 April 2009

Author's response to reviews:

referencing changes as requested in last email have been made
I have not made the other changes. For example, I am not aware that a table in journal articles requires more than one column
I can confirm that that I have submitted and had published articles in ISI listed journals where a table had one column